CREDIT Research Paper Centre for Research in Economic Development and International Trade, University of Nottingham



No. 02/20

The Volatility of Capital Inflows: Measures and Trends for Developing Countries

by

Robert Osei, Oliver Morrissey and Robert Lensink



The Centre for Research in Economic Development and International Trade is based in the School of Economics at the University of Nottingham. It aims to promote research in all aspects of economic development and international trade on both a long term and a short term basis. To this end, CREDIT organises seminar series on Development Economics, acts as a point for collaborative research with other UK and overseas institutions and publishes research papers on topics central to its interests. A list of CREDIT Research Papers is given on the final page of this publication.

Authors who wish to submit a paper for publication should send their manuscript to the Editor of the CREDIT Research Papers, Professor M F Bleaney, at:

Centre for Research in Economic Development and International Trade, School of Economics, University of Nottingham, University Park, Nottingham, NG7 2RD, UNITED KINGDOM

Telephone (0115) 951 5620 Fax: (0115) 951 4159

CREDIT Research Papers are distributed free of charge to members of the Centre. Enquiries concerning copies of individual Research Papers or CREDIT membership should be addressed to the CREDIT Secretary at the above address. Papers may also be downloaded from the School of Economics web site at: www.nottingham.ac.uk/economics/research/credit **CREDIT Research Paper** Centre for Research in Economic Development and International Trade, University of Nottingham



No. 02/20

The Volatility of Capital Inflows: Measures and Trends for Developing Countries

by

Robert Osei, Oliver Morrissey and Robert Lensink



The Authors

Robert Osei is a Research Officer in CREDIT and Oliver Morrissey is Director of CREDIT and Reader in Development Economics, School of Economics, University of Nottingham. Robert Lensink is Professor in the Faculty of Economics, University of Groningen, and External CREDIT Fellow.

Acknowledgements

This work was supported by a research grant from DFID (R7624) as part of the Globalisation and Poverty Research Programme (www.gapresearch.org). Useful comments on an earlier version were received from the Programme director, John Humphrey.

September 2002

The Volatility of Capital Inflows: Measures and Trends for Developing Countries

by Robert Osei, Oliver Morrissey and Robert Lensink

Abstract

This paper reports on trends and levels of capital inflows, and the volatility of such inflows, to a sample of 60 developing countries over the period from 1970 to 1997. The data cover aid and other development finance as the principal forms of official flows, FDI and other private flows, and debt (stock and flows). For each type of inflow to each country, three alternative measures of instability are calculated. To summarise the results, the countries are grouped into low income, lower middle and upper middle income. The measures of instability for each type of flow in each group and the evolution over time are discussed. The paper provides evidence that instability has increased in the 1990s (relative to the 1980s, but not to the 1970s), that official flows are less volatile than private flows, and the instability in FDI is lower than in other private flows. The paper also shows that the poorest countries have become increasingly reliant on aid and debt finance, attracting almost no private capital and little FDI. Only the richer developing countries attract significant volumes of FDI and private capital (but both are quite volatile).

Outline

- 1. Introduction
- 2. Measures of Volatility
- 3. Data and Trends in Composition of Capital Inflows
- 4. Trends in Volatility of Capital Inflows
- 5. Conclusions: Is Volatility a Cause for Concern?

1 INTRODUCTION

Foreign capital inflows are an essential source of investment finance for all countries, especially those on lower income levels (for which domestic savings rates tend to be low). Stable streams of inflows will support investment planning, whereas unstable flows may discourage investment and hence have a detrimental effect on growth (Lensink and Morrissey, 2000). Recently, some commentators have argued that increasingly volatile capital flows may lead to macroeconomic instability and contribute to financial crises (e.g. Gabriele *et al*, 2000). How volatile are capital inflows to developing countries and has this changed over time? The aim of this paper is to provide data on trends in the composition and volatility of capital flow, and their importance to specific countries differs. This paper concentrates on four types - debt, official flows (aid), foreign direct investment (FDI) and other private flows – and three groups of countries – low, lower middle and upper middle income. We consider trends in the importance and volatility of each type of inflow.

The core issue of concern is the degree of year-on-year variability of inflows – are the flows steady (or stable), or are they relatively volatile (or unstable). Although some writers have offered a distinction between volatility and instability (Gabriele et al 2000), the literature tends to use the terms interchangeably and we follow that tradition. The measures of instability (or volatility) most commonly used in the literature are deviations around a trend so that usually the measure can be interpreted as the percentage of mean (trend) value. The main difference between measures is the way in which the trend is estimated (a simple time trend or a forecast equation) and whether the measure is standard deviation around a trend or of a detrended series. Bleaney et al (1995), for example, measure revenue instability as the standard deviation of year on year changes for a detrended series. Gabriele et al (2000) use the coefficient of variation for the data in levels form to measure instability and the standard deviation of annual percentage change to measure volatility. Pallage and Robe (2002) measure aid instability as the percentage (standard) deviation around a detrended series (by removing the growth component so they can focus on the cyclical component).

A number of recent studies have measured instability but none have as comprehensive a coverage as we offer. Pallage and Robe (2002) consider only aid flows; for the sample of 38 African countries over 1969-95, instability of net aid receipts was 25% compared to 32% for commitments, whereas for (25) other developing countries instability was 30% and 40% respectively. Instability of multilateral aid is greater than for bilateral aid (both net receipts and commitments), while aid instability is considerably greater than output instability (13% on average for both the African and other developing countries samples). They also find, significantly, that aid tends to be counter-cyclical and therefore does not play a consumption smoothing role. Volatile capital flows can both reflect and exacerbate economic instability. Lensink and Morrissey (2000) argue that volatility of aid receipts is likely to be greater for economies that are vulnerable to shocks, while the associated unpredictability of aid receipts can undermine investment and budgetary management.

Twenty-five countries were chosen for the instability and volatility analysis, comparing the 1980s and 1990s, in Gabriele *et al* (2000). In many countries the volatility of capital flows increased in the 1990s, while it decreased in only a few (the authors report estimates for each country but no overall averages). Most African countries show a

greater degree of volatility in private capital inflows than Asian or Latin American countries in their sample. For FDI, both mean flows and standard deviations have gone up for most countries. Instability in FDI flows is noticeably greater for the latter period in absolute terms. Instability in net short-term flows increases in the 1990s with respect to the previous decade in eight countries out of thirteen. Instability appeared to increase in the countries studied.

Gabriele *et al* (2000) also show that capital flows to developing countries have increased substantially in recent years. Capital inflows are increasingly concentrated in a small group of emerging markets, while most developing countries continue to face a severe scarcity of external resources. The composition has shifted in favour of private flows: these were about two-thirds of the total in the 1970s, fell to 54 per cent in the 1980s, but rose thereafter until reaching about 80 per cent in the 1990s. Aid to developing countries reached a peak in relative terms in 1986 (at 25 per cent of inflows) then declined, falling below 10 per cent of total inflows from 1993. The composition of private flows also changed - bank loans declined while both FDI (which is highly concentrated in a small number of countries) and portfolio inflows increased.

The volume of capital flows to sub-Saharan Africa (SSA) also increased in the 1990s, although estimates vary according to the source of data (Bhinda *et al*, 1999). There is a tendency for international organisations to underestimate the scale of capital flows given difficulties in collecting and assembling comparable data from national sources. Although only a small fraction of global FDI, FDI to SSA tripled between 1992 and 1995, growing fastest in some of the least developed economies. The IMF and World Bank report that portfolio equity inflows are also rising fastest in SSA among developing country regions, albeit from a very low base. The level of net inflows through bank loans has decreased in SSA, although net short-term flows have risen in some countries, such as Tanzania since 1996 (Bhinda *et al*, 1999).

We expand on the evidence from these studies by considering a variety of types of flows for a large number of developing countries since the 1970s (the largest and longest sample we could compile). Section 2 describes the three measures of volatility we employ in our analysis. Our concern is to quantify how variable each type of capital flow is to the three groups of developing countries. Thus, it is not important whether this is described as volatility or instability. Section 3 presents an overview of trends in capital flows to developing countries, identifying how the composition has changed over the past three decades. Section 4 then presents the results on instability, considering also the trends in instability and correlation with growth. Section 5 presents a summary of the findings.

2 MEASURES OF VOLATILITY

We compute and compare three different measures of instability. The first is the coefficient of variation (CoV) of the series, the standard deviation expressed as a percentage of the mean value (for the series over time). The second (Index I) is the standard deviation around a simple time trend. The third measure (Index II) is the standard deviation around a forecast trend. The forecast is based on adaptive expectations such that in principle it captures the trend value that would be predicted using past values as a guide. There is no strong reason to favour one index over the other. If one believes that a planner (government or investor) bases expectations on a simple time trend, index I is appropriate. If the planner uses past values to form a

forecast, index II is appropriate. If both give similar values and/or trends, we can judge how volatile flows appear to the country.

Index I

Following Cnossen and Morrissey (1997) we calculate this as the normalised standard deviation of the residuals from a time trend.¹ For a given capital flow y we estimate the trend equation as:

$$y_t = \alpha + \beta T + \varepsilon_t \tag{1}$$

where α and β are parameters, *T* is the time trend, and ε_t is the deviation of the actual series from the linear time trend. Given that the number of parameters to be estimated is two the minimum period average one can calculate for the index is three (which happens to maximise the degrees of freedom for the annual series in this case). The index is then calculated as (where \overline{y} is the arithmetic mean of *y*):

$$\frac{100}{\overline{y}}\sqrt{\frac{1}{n-3}\sum_{t=1}^{n}\varepsilon_{t}^{2}}$$
(2)

Index II

This is calculated as normalised standard deviation of residuals from the forecast (expected) values. This is quite similar to the first index except that the deviations are from some expected (forecast) values (from an autoregressive model). It is calculated as (where $\mu_t = y - \hat{y}$ and \hat{y} is estimated from a simple AR(*p*) model):

$$\frac{100}{\overline{y}}\sqrt{\frac{l}{n-p}\sum_{t=1}^{n}\mu_t^2}$$
(3)

Given that most of the series are non-stationary we estimate the model in first differences and then recalculate the forecast values in levels to make it comparable to index I. This we do by following these simple steps. For an I(1) series (which is true for majority of the series) we calculate an autoregressive process (of order 3) by estimating the equation:

$$\Delta y_t = \eta_0 + \eta_1 \Delta y_{t-1} + \eta_2 \Delta y_{t-2} + \kappa_t \tag{4}$$

Having obtained estimates for η_t 's we can calculate the forecast values of y_t by rewriting the equation as

$$\hat{y}_{t} - y_{t-1} = \hat{\eta}_{0} + \hat{\eta}_{1}(y_{t-1} - y_{t-2}) + \hat{\eta}_{2}(y_{t-2} - y_{t-3})$$
(5)

Grouping terms we get the level forecast of y_t as

$$\hat{y}_{t} = \hat{\eta}_{0} + (\hat{\eta}_{1} + 1)y_{t-1} + (\hat{\eta}_{2} - \hat{\eta}_{1})y_{t-2} - \hat{\eta}_{2}y_{t-3}$$
(6)

¹ Although a quadratic trend is used in that study, our initial estimates suggest that a quadratic time trend does not improve the fit.

We now obtain the μ_t as the difference between the actual series and the level forecast values from equation (6).

3 DATA AND TRENDS IN COMPOSITION OF CAPITAL INFLOWS

The developing countries in our sample are classified into low, lower-middle, and upper-middle income countries (LIC, LMIC, and UMIC respectively). The LICs are economies in which per capita GNP in 1998 was at most \$760 (there are 29 countries in this group). The LMICs (of which there are 19) and UMICs (12 countries) are those economies in which per capita GNP in 1998 was within the ranges \$760 to \$3,030 and \$3,031 to \$9,360 respectively. The tables in Appendix A list the countries under each group. In general the period under study is from 1970 to 1997. All the capital flows and external debt data are expressed as a per cent of GDP; given the wide variety of countries being considered, this is the most simple means to provide a measure of the 'real' value of the inflow. As our aim is to evaluate the importance and instability of the inflow from the perspective of the receiving country, this is quite appropriate.

The variables included in the study are net disbursements of Official Development Assistance (ODA), net disbursements of Official Development Finance (ODF), Foreign Direct Investment (FDI), Other Private Capital flows (OPC) and total External debt (Debt, measured as both stock and flows). Full definitions and sources are provided in Appendix A. There are thus two measures of aid flows, ODA which represents official concessional finance and ODF which also includes certain non-concessional official finance (thus, ODA is a sub-set of ODF). There are also two measures of private flows, FDI and OPC (mostly commercial bank lending, bonds, and portfolio equity investments). There are also two debt measures, the stock and the flow (change in the stock). Tables 1a-1c present the trends in each capital inflow to each group of developing countries (the figures are the ratio of the flow to GDP on average for the countries in each group), summarised in Figure 1.

	Debt (stock)	Debt (flows)	FDI	OPC	ODA	ODF
1970 - 1975	23.56	4.49	0.97	1.06	6.35	6.58
1976 - 1980	37.88	7.98	1.04	2.09	8.28	9.18
1981 - 1985	63.57	5.94	0.47	0.98	8.89	9.97
1986 - 1990	99.16	9.15	0.51	0.07	12.00	12.89
1991 – 1995	123.63	3.75	0.82	0.03	14.50	14.83
1996 - 1997	108.34	-6.11	1.44	0.17	12.21	12.61
Entire period	72.43	5.40	0.82	0.81	10.03	10.68
St. Deviation	39.18	4.98	0.39	0.91	3.20	3.20
Minimum	21.55	-9.20	-0.10	-0.30	5.10	5.53
Maximum	143.16	15.22	1.55	2.67	18.97	19.01

Table 1a: Trends in Capital Inflows (ratios of GDP) - LICs

Notes: Figures are the mean for all LICs of the average annual value of each capital flow for each period. Minimum and maximum thus refer to a year.

	Debt (stock)	Debt (flows)	FDI	OPC	ODA	ODF
1970 – 1975	27.58	3.88	1.20	1.35	3.23	3.56
1976 – 1980	41.91	8.21	1.13	2.35	4.49	5.23
1981 – 1985	63.89	6.39	0.84	1.44	4.03	5.12
1986 – 1990	75.84	3.61	1.32	0.54	4.18	5.04
1991 – 1995	61.90	1.75	1.93	0.15	3.40	4.21
1996 – 1997	52.28	0.35	2.62	1.29	2.09	2.80
Entire period	54.09	4.46	1.38	1.18	3.72	4.46
St. Deviation	17.72	3.58	0.57	0.94	0.89	1.00
Minimum	26.40	-2.37	0.57	-0.49	1.91	2.44
Maximum	79.47	12.99	3.20	3.64	5.20	5.98

Table 1b: Trends in Capital Inflows (ratios of GDP) - LMICs

Notes: Figures are the mean for all LMICs of the average annual value of each capital flow for each period. Minimum and maximum thus refer to a year.

	Debt (stock)	Debt (flows)	FDI	OPC	ODA	ODF
1970 – 1975	22.60	5.32	1.70	2.23	2.42	3.25
1976 – 1980	32.69	6.27	1.56	3.15	1.62	2.00
1981 – 1985	46.16	3.82	1.29	1.87	1.49	2.89
1986 – 1990	56.35	3.14	1.34	0.45	1.22	2.27
1991 – 1995	47.32	1.87	1.88	1.17	0.77	1.10
1996 - 1997	43.74	1.32	3.79	3.42	0.52	0.86
Entire Period	41.23	3.82	1.72	1.91	1.47	2.13
St. Deviation	12.37	2.45	0.80	1.20	0.65	1.00
Minimum	19.25	-0.48	0.93	-0.10	0.51	0.90
Maximum	61.79	8.24	4.88	3.99	2.87	5.67

Table 1c: Trends in Capital Inflows (ratios of GDP) – UMICs

Notes: Figures are the mean for all UMICs of the average annual value of each capital flow for each period. Minimum and maximum thus refer to a year.





Trends in Capital Flows by Country Groups, 1970s to 1990s

8

The LICs are the only grouping for which total inflows have increased (relative to GDP) over the period; aid has become more important and debt flows less important (Figure 1). For the LICs (Table 1a), both net ODA and ODF flows have followed a similar trend (as would be expected give that non-aid ODF is quite small for the poorest countries, less than ten per cent of the total as can be seen from the difference between ODF and ODA). Net ODA averaged about 10 per cent of GDP over the entire period with a standard deviation of about 3.2 (implying a simple CoV of 32%). Overall it has shown a steady increase from about six per cent of GDP in 1970-75, peaking at almost 15% 1991–95 before decreasing slightly to about 12% in 1996–97. The trend is very similar for ODF and also when one looks at gross ODA which averaged about 11 per cent over the period with a standard deviation of about 3.6 per cent (see Appendix).

Compared to official flows, private capital flows for the LICs have remained at very low levels over the entire period. The entire period averages for both FDI and OPC were less than one per cent of GDP, with standard deviations of about 0.4 and 0.9 respectively. The implied simple CoVs support the common perception that FDI is less volatile than OPC. Looking at private capital flows over time one observes that in general the trend has been negative. For instance, FDI decreased from about one per cent over 1970-80 to about 0.5 per cent in the 1980s, although it then increased to reach a peak of 1.4 per cent in 1996–97. OPC, on the other hand, decreased fairly steadily from two per cent in the late 1970s to almost zero in the early 1990s. Total private capital flows have declined by about a third over the whole period, with OPC becoming less important relative to FDI.

Total external debt ratios for the LICs are quite high – the entire period average is about 72 and five per cent for the stock and flows respectively. In other words, debt flows have on average increased by over five per cent every year for the LICs. Looking at the stock of debt one observes that it has not always been high. Most of the increase occurred over the post-1980 period. It increased from a period average of about 24% over 1970–75 to well over 100% by the 1990s. Over the last two years of the sample the stock of debt has decreased, as shown by the negative average debt flows in 1996-97. Volatility of debt is high, as can be seen from the relatively high standard deviations.

Total flows to LMICs relative to GDP declined from 13% in the 1970s to eight per cent in the 1990s; aid has become relatively more important, OPC has declined, although FDI has become significant (Figure 1). Both net ODA and ODF flows (as a per cent of GDP) seem to have increased over the 1980s but subsequently declined in the 1990s (Table 1b). Average net ODA and ODF over the sample period were about 3.7 and 4.5 per cent respectively for the LMICs. Thus, non-aid official flows are clearly more important than for LICs (amounting to about 17% of ODF for LMICs). From a period average of about 3.2% in 1970–75, net ODA increased to four per cent in 1986–90 before declining to two per cent by 1996–97. Average gross ODA flows over the sample period were about 4.2% for the LMICs (Appendix Table A4a). The pattern for ODF is similar to ODA; increasing from about 3.6% in 1970–75 to over five per cent in the 1980s before decreasing to less than three per cent by 1996–97.

Private capital flows (as a per cent of GDP) have not changed much over the years. On average FDI and OPC were about 1.4 and 1.2 per cent respectively, and again OPC was considerably more volatile (as indicated by the higher standard deviation). Period average FDI more than doubled from 1.2% in 1970–75 to 2.6% in 1996–97, with a decline in the early 1980s. OPC increased in the late 1970s, then declined, quite

dramatically by the early 1980s, but ended the period in 1996-97 at a comparable level to the early 1970s (about 1.3% of GDP).

Total external debt of the LMICs averaged 54% of GDP over the entire period. There was a steady increase from almost 30% in the early 1970s to 75% by the late 1980s before declining to 52% in 1996–97. This trend is reinforced by looking at debt flows; although over the entire period the average increase was some 4.5%, this was heavily concentrated in the 1980s and the rate of increase fell dramatically in the 1990s. It is also evident that volatility is high for debt, especially for flows. Thus, whereas debt burdens were rising for the LICs, they have (at least recently) been falling for LMICs.

Total flows to UMICs fell by almost half relative to GDP over the period, with aid of declining importance and OPC and especially FDI of increasing importance (Figure 1). As would be expected, official flows to the UMICs have decreased over the years (Table 1c). Net ODA (as a per cent of GDP) decreased from 2.4% per cent to about 0.5% between the early 1970s and the late 1990s. Although significantly larger (the non-aid share is almost 30%), ODF also decreased from over three per cent to less than one per cent over the same period. To a large extent the decrease in official flows have been compensated for by an increase in private capital flows for this group of countries. The entire period averages for private capital flows are 1.7 and 1.9 per cent for FDI and OPC respectively. FDI more than doubled from 1.7% in the early 1970s to 3.8% in the late 1990s. Although much more variable (and with greater volatility than FDI), OPC also increased from about 2.2% in the early 1970s to 3.4% in the late 1990s, albeit with a large dip in the late 1980s.

Debt levels appear far more manageable for UMICS. External debt averaged about 41 per cent of GDP over the entire period and only over the 1986–90 period did it exceed 50 per cent. In general the trend has been an increasing one – from over 20% in the early 1970s to over 40% by the late 1990s. Debt flows exhibit this steady increase, averaging almost four per cent although it is evident that the rate of increase has declined steadily.

Overall, the trends confirm what would have been expected *a priori*. Official flows, especially aid, are most important for poorer countries and have become of increasing importance over time. Private capital flows to LICs have declined (relative to GDP) – although the decline in FDI has been reversed in the 1990s this was insufficient to off-set the collapse in other private capital inflows. For LMICs, private inflows have increased, although all of the increase is due to FDI (OPC has remained steady overall). Only UMICs have experienced an increase in OPC, although all of this was in the 1990s and the increase in FDI was greater. Furthermore, we can infer that LICS experienced a much poorer growth performance than richer developing countries. The average figures for debt flows are not dramatically different for the country groups, but debt burdens rose dramatically in LICs (by more than four times), increased by more than half in LMICs but almost doubled in UMICs. On a more positive note, debt burdens declined for all groups in the late 1990s.

4 TRENDS IN VOLATILITY OF CAPITAL INFLOWS

While the data discussed in the previous section identifies the trends of capital inflows over the past three decades, and we have seen that these are variable, we are here concerned with how volatile or unstable the flows have been. That is, we want measures that capture the year on year variability of the flow, and that summarise this volatility over the whole period (and allow for a decomposition for sub-periods). Three measures of instability were proposed in Section 2. All indices are normalised so that the value can be interpreted as a percentage of the mean. The overall (1970-97 period) values for flows to each group of countries are given in Table 2 (the figures are instability of the average ratio of the flow to GDP for the countries in each group). Instability values for individual countries are in Appendix Tables A1a-A3c, and tend to be much higher than instability of average flows to the countries as a group. We highlight the main features of the individual country results in the discussion.

	Debt (stock)	Debt	FDI	OPC	ODA	ODF
		(flows)	LICS	5		
CoV	54.09	94.37	47.68	113.14	31.89	29.98
Index I	14.94	91.21	48.52	93.51	14.47	13.39
Index II	11.14	96.39 44.59 69.54 17.85 1 <i>LMICs</i>				
CoV	32.77	80.23	41.64	79.73	24.01	22.37
Index I	23.92	70.61	33.14	72.07	24.15	22.79
Index II	8.17	81.78	29.74 <i>UMI</i> C	60.25 Cs	14.68	13.63
CoV	30.00	64.13	46.24	62.99	44.02	46.56
Index I	19.87	48.33	44.36	60.82	14.50	35.87
Index II	8.77	59.88	38.80	46.36	13.36	22.60

Table 2: Instability of Capital Flows (entire period, flows to all countries)

Notes: Estimates based on instability of average flows to the countries in each group. Instability measures are defined in Section 2. Estimates for each individual country are presented in Appendix Tables A1a-A1c.

Official Flows (aid and ODF)

For the LICs aid (ODA) is slightly more volatile than ODF and for both CoV is much higher than either index, implying that much of the overall variation is due to trends. Surprisingly, perhaps, Index I is lower than Index II implying that the AR forecast picks up less of the variability (i.e. performs less accurately) than the simple time trend. In most cases in Table 2, however, Index I has a higher value than Index II, as would be expected if the AR 'forecast' is somewhat 'better' than a simple time trend. The distinction between them (for overall averages and trends) is not so important, and we interpret them as providing a range of instability. Thus, instability of aid flows to LICs is 14-18% (of the mean?). These official flows are considerably less volatile than private flows.

Instability measures for the 29 individual LICs (over the entire sample) are given in Appendix Table A1a. The modal value of ODA instability indices (I and II) is about 35% (seven countries are in the 30-40% range for both indices, a further six for one index). This is much higher than the period values in Table 2 because aid to individual countries will be more volatile than average aid to LICs. Nine countries had values lower than 30% (both indices) while six had values greater than 50% (both indices). Rwanda, Nigeria, Cote d'Ivoire, Nicaragua and Haiti are the five countries that exhibit the highest values for the instability indices for official flows. In general, in the LIC sample, instability of official flows seems to be higher for the African countries. Almost three-quarters of the countries with an index value greater than 50 per cent are African countries.

Overall instability of aid flows to LMICs is 24% for Index I and 15% for Index II (so a forecast value seems more reliable than a simple trend). The values for ODF are similar (Table 2). Thus, volatility of official flows is similar to that for LICs and is lower than volatility of private flows. Average instability in ODA for the 19 LMICs is about 50% (Table A1b). For most countries Index II is significantly lower than Index I implying that, in principle, much of the overall instability could be anticipated. The modal value of Index I is 44% (six countries are in the range 40-48%), with five countries below 30% and eight above 50%. The median value of Index II is 35% (seven countries in range 30-39%), with seven countries below 30% and four above 50%. Instability in ODA is highest (using Index II) for Dominican Republic, Costa Rica, Morocco and Jordan. The list of countries varies if one chooses Index I or ODF, although the broad pattern is similar.

Of the three groups of countries, instability for UMICs is lowest for ODA at 13-15% but highest for ODF at 23-36% (Table 2). Instability of ODF is much higher than of ODA, so it is non-aid flows that are most volatile (and note that non-aid official flows are more important for UMICs). Index II is considerably lower than Index I, implying that more than a third of instability can be captured by the forecast as compared to the trend estimate. Average instability for the 12 UMICs is 64% and 86% for ODA and ODF respectively for Index 1. The modal value for Index II is about 45% and 55% for ODA and ODF are Chile, Venezuela, Trinidad & Tobago, Turkey and Malaysia – these are also the only countries with instability greater than 50%. The ranking differs for ODF, although Chile, Venezuela, Trinidad & Tobago, are the three with the highest instability. It is evident, more so than for other country groups, that although instability of aid to all UMICs is quite low, instability of flows to individual countries is very high. This also applies for the higher levels of instability observed for ODF.

Private Capital flows

Instability in private capital flows has been much higher than in official flows for all three groups of countries and FDI is less volatile than OPC (Table 2). Instability of total FDI to LICs was 45%, compared to 70% for OPC (Index II). Instability of total flows to LMICs was lower, 30% for FDI and 60% for OPC. Instability of flows to UMICs was greater for FDI at 39% but lower for OPC at 46%. In all cases, Index II gave lower instability than Index I, often considerably so (implying that the AR process is a better representation than a simple time trend). Levels of instability for individual countries are, however, much higher.

For individual LICs, modal instability (Index II) in FDI is about 130% (nine countries in range 100-160%); 13 countries have values less than 100% (eight less than 60%) and seven have values greater than 200% (Table A2a). Modal instability (Index II) in OPC is about 200% (nine countries are in the range 100-200% and five in the range 200-300%); only three countries have values less than 100% and six have values greater than 500% (Table A2a). The five countries with the highest instability for FDI (Index II) are all African, namely Mauritania, Sierra Leone, Congo DR, Sudan and Togo. The seven countries with Index I greater than 200% are African. Looking at OPC, the six countries with Index II above 500% are Nigeria, Solomon Islands, Gambia, Haiti, Mauritania and Rwanda. About four-fifths of the countries with Index I greater than 200% are African.

For individual LMICs, modal instability (Index II) in FDI is about 70% (five countries in 50-90% range); four countries have values less than 50% and five have values greater than 100% (Table A2b). Algeria, Jordan, Peru, Bolivia and El Salvador are the five countries with the highest instability for FDI and the only ones with Index II above 100%. Modal instability (Index II) in OPC is about 115% (nine countries are in the range 100-200% but five of these are within 109-124%); four countries have values within 90-100% and four have values greater than 500% (Table A2b). Jamaica and Swaziland have very high negative values as on average there was an outflow of net OPC from these countries (respectively –0.05 and –0.18 per cent annual rates). The five countries with the highest (positive) Index II for OPC are El Salvador, Sri Lanka, Guatemala, Tunisia and Paraguay.

Average instability in FDI and OPC for the UMICs is respectively 109% and 272% using Index I. Modal instability for Index II is about 75% for FDI (five countries below this with five above but less than 200%), with only Venezuela and Botswana above 200%. Modal instability for OPC is about 100% (seven countries within 50-150%). Trinidad & Tobago (at a huge 1611%), Botswana, Barbados, Venezuela and Mauritius exhibit the highest values and are the only countries with instability exceeding 200%. In general, instability in private capital flows is lower for this group of countries than for the LICs, but much higher than for official flows.

External Debt – Stock and Flows

Volatility seems to be much higher in debt flows than it is in the stock of debt, as would be expected, debt flows are the most volatile capital inflows for all three groups of countries (Table 2). Instability of debt flows to LICs was 96% (Index II), compared to 82% for LMICs and 60% for UMICs. In all cases, Index II gave higher instability than Index I, and CoV was often quite similar. This suggests that net debt flows (or, more strictly, the annual change in the debt stock), follows an unpredictable and volatile path.

For individual LICs, instability in debt flows has a modal value around 120% (Index II), with seven countries below 100%, twelve within 100-150% and seven above 200%. The five countries with the highest level of instability are Nicaragua, Madagascar, Sierra Leone, Niger and Togo. Modal instability for debt flows of individual LMICs is about 120%, with six countries below 100% and six above 150%. The five countries with the highest instability are Swaziland, Egypt, Jamaica, Jordan and Paraguay. Among the UMICs the modal value is about 120%, with two countries below 100% and three above 150%. Barbados, Trinidad & Tobago and Venezuela had the highest levels of instability.

Trends in Instability

Our discussion on the trends in instability is based on the indices calculated for capital flows to all countries in each of the three country groups. The trends are based on the measure of annual instability calculated as the first difference of the recursive series (see Section 2). Table 3 presents the annual average change in instability for Index II for all flows over the whole period and three sub-periods, the 1970s, 1980s and 1990s. Although the values differ, the trends are broadly similar for Index I (Appendix Table A5). In general, instability has been declining over the whole period (the yearly average is negative in most cases). This is somewhat misleading as, for most flows, instability declined significantly in the 1970s but has increased since then; this is illustrated in Figures 2-4 for aid, FDI and OPC.

	Debt (stock)	Debt (Flows)	FDI	OPC	ODA	ODF
		LIC	Ċs			
1972 – 1980	-4.32	-25.22	-5.60	-8.72	-2.72	-2.67
1981 – 1989	0.44	2.32	0.68	0.97	-0.41	-0.39
1990 – 1997	-0.30	1.67	-1.24	1.97	1.02	0.91
Period	-0.57	-0.85	-1.25	-0.50	-0.31	-0.33
		LMI	Cs			
1972 – 1980	-3.87	-24.67	-4.03	-	-2.06	-2.01
				11.26		
1981 – 1989	-0.64	-1.17	-0.06	-1.13	-0.54	-0.64
1990 – 1997	-0.12	1.28	-0.69	0.96	0.28	0.32
Period	-0.92	-2.61	-0.91	-1.81	-0.44	-0.46
		UMI	Cs			
1972 – 1980	-5.01	-8.64	-0.89	-1.89	-2.38	-3.18
1981 – 1989	-0.39	1.82	-0.62	1.11	-0.42	-0.49
1990 – 1997	-0.15	0.09	1.07	-0.03	-0.02	0.12
Period	-0.99	-0.01	-0.58	-0.71	-0.93	-1.19

Table 3: Instability of Capital Flows (Index II, annual average trend)

Notes: Estimates based on average flows to countries in each group. Annual instability measures are obtained as the first difference of the recursive estimates for Index II. Period is annual average over 1970-97. Trends for Index I are in Appendix Table A5.

Figure 2 Annual change in average instability (Index II) – ODA



Notes: Compare with Table 3 for period values of instability (figure shows change not level).





Notes: As for Figure 2.





The trends for instability of inflows to LICs are in the top panel of Table 3. Instability in aid fell in the 1970s but has increased significantly in the 1990s; the spike in 1994 is evident in Figure 2. Instability of OPC fell dramatically in the 1970s (implying that flows were very volatile in that period), but have increased since then (especially in the 1990s, see Figure 4). FDI has been variable, and is the only flow for which instability declined n the 1990s (Figure 3). The instability of external debt flows fell dramatically in the 1970s but has increased steadily since then, especially in the 1980s. Overall, instability has been increasing in the 1990s, especially for aid, OPC and debt flows.

The broad trends are similar for LMICs, except that for most flows the large reduction in instability in the 1970s was followed by a further slight reduction in the 1980s, then instability began to rise in the 1990s. The exception is FDI, instability of which continued to decline in the 1990s (Figure 3). As for LICs, instability has been rising in the 1990s (but only slightly, except for debt flows), except in the case of FDI. UMICs exhibit a similar pattern but the changes have been more moderate. Following moderate declines in instability of private and debt flows in the 1970s, there have been steady increases in the 1980s and 1990s (especially for OPC). Official flows saw declining instability until the 1990s, when instability remained almost unchanged. Only in the case of FDI has there been a significant increase in instability in the 1990s (Figure 3).

Volatility of Capital Inflows and Economic Performance

For all three groups of countries we observe that private capital and external debt flows have exhibited the highest volatility, while FDI and, especially, official flows have the lowest instability. Although there seems to have been a general decrease in volatility over the whole period, this occurred almost completely in the 1970s and the data suggest that instability in capital flows has increased in the 1990s. Furthermore, the rate of increase has been highest in the most volatile flows, OPC. Table 3 suggests that volatility was greatest (in the sense that trend changes were very high) but declining in the 1970s, and has increased in the 1990s. We now consider if instability for individual countries exhibits any correlation with growth.

Tables 4a-4c give, across the countries in each group, the correlation between instability of capital flows (measured using Index I), output growth (GDP growth rates) and the Commonwealth Vulnerability Index (CVI). The CVI quantifies a country's relative economic and ecological susceptibility to exogenous shocks. It has two components – the impact component and resilience. The impact component reflects the incidence and intensity of risk and threats to a country. It is proxied by the expected value of income volatility having controlled for economic exposure, remoteness and insularity, and susceptibility to environmental events and hazards. Resilience reflects the ability of a country to insulate itself from the risks and threats and to bounce back from external economic and environmental shocks. It is proxied by a country's 1995 GDP. A weighted combination of these two components forms the CVI. The data are obtained from Atkins *et al* (2000).

For the LICs, GDP growth is negatively correlated with volatility in all the variables except OPC, for which the coefficient is anyway very low (Table 4a). These correlation coefficients suggest that LICs with relatively low growth rates are those that have experienced relatively high volatility in financial flows. FDI volatility is strongly correlated with GDP growth with a (negative) coefficient of over 40 per cent, which may be because countries with poor growth are less attractive to FDI. There is also a high

negative correlation between growth and instability of debt flows (and debt instability measures are quite highly positively correlated with instability of other inflows, except FDI). Lower growth is associated with higher volatility of debt flows.

The correlation between CVI and volatility in financial flows (except for ODF) are in general positive but very small. This suggests that LICs that are relatively more vulnerable are those that have relatively high volatility in their financial flows, although this is only true at a moderately significant level (20%) for OPC. We note a positive, albeit low (18%), correlation between CVI and growth. This may seem counter-intuitive as one would expect countries that are vulnerable to have lower growth rates. However this positive correlation can be explained when one considers the components of the CVI in relation to GDP growth. First the impact component of CVI is measured as income volatility (the standard deviation of annual rates of growth of per capita GDP). The impact component could very well be positively correlated with GDP growth. The second component is resilience (1995 GDP) and there is no reason why one should expect this to be negatively correlated with real GDP growth.

	GDP Growth	CVI	Debt (Stock)	Debt (flows)
FDI	-0.4039	0.0035	-0.1228	0.0221
ODA	-0.0593	0.0025	0.5804	0.3056
ODF	-0.1463	-0.0252	0.6377	0.3482
OPC	0.0441	0.2048	0.3054	0.1404
Debt Stock	-0.1915	0.1237		
Debt flows	-0.4392	0.0074		

Table 4a: Correlation between Instability, Vulnerability and Growth – LICs

Notes: Simple correlation coefficients. Instability measure is Index I for all flows, and CVI is the Commonwealth Vulnerability Index (see text). Correlation between CVI and GDP growth is 0.184.

	GDP Growth	CVI	Debt (Stock)	Debt (flows)
FDI	-0.2713	-0.0635	-0.1461	0.2156
ODA	-0.1130	-0.1325	0.1491	0.0370
ODF	-0.0865	-0.0107	-0.0132	0.1525
OPC	0.4666	-0.1546	0.0286	-0.4174
Debt Stock	-0.3388	0.1883		
Debt Flows	-0.0781	0.6634		

Table 4b: Correlation between Instability, Vulnerability and Growth – LMICs

Notes: As for Table 4a. Correlation between CVI and GDP growth is 0.147.

Table 4c: Correlation betv	veen Instability,	Vulnerability	and Growth	– UMICs
		•		

	GDP Growth	CVI	Debt (Stock)	Debt (flows)
FDI	0.1413	0.4740	-0.1907	0.3619
ODA	-0.4296	-0.6696	0.1972	-0.0696
ODF	-0.3183	-0.3133	0.1390	0.1868
OPC	-0.1175	0.2093	-0.1292	0.5972
Debt stock	0.1064	0.0197		
Debt Flows	-0.2552	0.4841		

Notes: As for Table 4a. Correlation between CVI and GDP growth is 0.498.

For LMICs, GDP growth is most strongly and positively correlated with OPC volatility (47%, Table 4b). This is consistent with a view that private capital is responsive, but very sensitive, to economic growth – it flows in to better performing countries, but can flow out if there are small changes in performance. We note a relatively high negative correlation between volatility of OPC and debt flows, implying that countries with high OPC instability are not the same countries that have volatile debt flows. Volatility in the other flows and debt exhibit a negative correlation with GDP growth, although this is very low except for FDI and debt stock. Countries with lower growth have higher instability in FDI and debt burdens. The correlations between CVI and volatility in both private and official capital flows are negative and fairly small, although volatility of debt flows is highly positively correlated with CVI; more vulnerable LMICs have more volatile debt flows. For this group of countries we note that CVI is negatively correlated with instability in financial flows but positively correlated with GDP growth.

Volatility in financial flows, with the exception of FDI, is negatively correlated with GDP growth for UMICs (Table 4c). The highest negative correlation is between ODA (and ODF) volatility and growth, perhaps because the best performing UMICs receive very low, albeit stable, official inflows. GDP growth also has a negative correlation with volatility in debt flows. As for other groups, growth is negatively correlated with OPC instability and positively with FDI instability; FDI and OPC appear to be directed at different sets of countries. There is quite a high positive correlation between CVI and growth, most likely for the reasons outlined above. There is a high negative correlation between CVI and volatility in official flows (although we note that such inflows are relatively minor to these countries). There is quite a high positive correlation between CVI and instability of FDI and debt flows. For this group of countries vulnerability is associated with volatility of capital inflows.

5 CONCLUSIONS: IS VOLATILITY A CAUSE FOR CONCERN?

This paper considers trends in the importance and volatility of four types of capital flows - debt, official flows (aid and ODF), foreign direct investment (FDI) and other private flows(OPC) – to three groups of countries – low (LIC), lower middle (LMIC) and upper middle (UMIC) income. Three measures of instability are proposed in Section 2. The coefficient of variation (CoV) is the standard deviation as a percentage of the mean over the period. This is a simple measure of total instability, but our ultimate concern is with what instability means to countries receiving capital inflows. Countries will have expectations about inflows, and may make macroeconomic plans according to such expectations, so we want a measure that tries to account for this. Two measures are provided. Index I measures instability as year on year deviations from a time trend, assuming that the expectation is that the inflows follows such a trend. Index II is the trend). Both indices are normalised so that the value can be interpreted as a percentage of the mean. In the summary below, rather than exploring the differences between the indices we report a rough average of the two indices as the measure of instability.

For the LICs in our sample, official flows have become relatively more important compared to private capital flows. Official inflows (aid) have increased from an average of six per cent of GDP in the early 1970s to an average of 12% in the late 1990s. While the FDI/GDP ratio has risen slightly (from about one to 1.4%), other private capital flows have fallen to negligible levels and debt/GDP ratios have risen from less than 25% to over 100%. The data seem to suggest that the high external debt ratios for LICs may be a

deterrent to private capital flows, as private inflows have slumped while debt burdens increased. The clear trend is for LICs to have become increasingly dependent on aid and debt finance (although both have declined in the late 1990s), with small but increasing FDI and negligible private capital inflows.

The LMICs exhibit an increasing reliance on official inflows and FDI, but a decline in private capital inflows (as a share of GDP). However, private flows are low: FDI only exceeded two per cent of GDP on average in the 1990s while private capital inflows declined significantly in the late 1980s and early 1990s, only recovering to over one per cent of GDP in the late 1990s. Thus, official flows are still more important than private capital. Although debt burdens increased significantly between the 1970s and late 1980s, they have subsequently been reduced and stood at an average of 52% of GDP in the late 1990s (compared to 28% in the early 1970s).

While official inflows have declined to less than one per cent of GDP on average for the UMICs and private inflows have increased, other flows have followed trends similar to the other groups. Both FDI and private capital flows declined in the 1980s but both recovered and have been more important than official flows in the 1990s. By the late 1990s, private capital inflows exceeded three per cent of GDP on average and FDI was almost four per cent of GDP on average. External debt burdens more than doubled in the 1980s (over 50% compared to 23% in the 1970s) but have been reduced to an average of below 45 per cent of GDP by the late 1990s.

All capital inflows exhibit instability over 1972-97, ranging from around 14% for aid in most cases to as high as 90% for debt flows to LICs. Instability in official flows is moderate for all three groups of countries on average (15-20%) and aid is more volatile than ODF except for flows to UMICs. Instability in private capital flows has been much higher than in official flows for all three groups of countries and FDI is less volatile than OPC. Instability of total FDI to LICs was 47%, compared to 82% for OPC. Instability of total flows to UMICs was lower, 32% for FDI and 66% for OPC. Instability of flows to UMICs was greater for FDI at 42% but lower for OPC at 54%. In all cases, Index II gave lower instability than Index I, often considerably so. In other words, deviations around a simple time trend tend to exaggerate volatility compared to volatility around a forecast trend.

Debt flows are the most volatile capital inflows for all three groups of countries. Instability of debt flows to LICs on average was 94%, compared to 76% for LMICs and 54% for UMICs. In all cases, Index II gave higher instability than Index I, and CoV was often quite similar. This suggests that net debt flows (or, more strictly, the annual change in the debt stock), follows an unpredictable and volatile path.

For most flows to all three groups of countries, instability fell considerably in the 1970s but has increased since then. Aid to LICs and debt flows to UMICS are the only cases where instability exhibited a trend increase over the entire (1972-97) period. In all other cases, the trend was for decreasing instability over the whole period, but the decrease was concentrated in the 1970s. In the case of LICs, instability has been increasing for all flows except FDI in the 1990s, especially for OPC and debt flows. The same pattern is observed for LMICs, where again FDI was the only flow for which instability declined in the 1990s. All flows to UMICs exhibit increasing instability in the 1990s (and the 1980s for non-official inflows), although in most cases the changes have been more moderate.

We find that LICs with relatively low growth rates have experienced relatively high volatility in financial flows. FDI volatility is strongly negatively correlated with GDP growth, possibly because countries with poor growth are less attractive to FDI. For LMICs, GDP growth is most strongly and positively correlated with OPC volatility. This is consistent with a view that private capital is responsive, but very sensitive, to economic growth. Countries with lower growth have higher instability in FDI and debt burdens. As for other groups, growth in UMICs is negatively correlated with OPC instability and positively with FDI instability; FDI and OPC appear to be directed at different sets of countries.

Overall, the composition of capital flows to developing countries has indeed changed, although starting from the 1970s there have been two phases: increasing debt burdens and declining private (FDI and other capital) flows in the 1980s, with a reduction in debt burdens and increasing capital inflows in the 1990s. These phases can be seen for all three groups of countries although the decline in the 1980s was most pronounced for LICs (spiralling debt and almost disappearing private inflows), and they have shown the least recovery. By the 1990s, poorer developing countries are reliant on aid for almost 80 per cent of capital inflows (debt financing contributes almost 15%); only the richer developing countries are attracting private capital inflows and FDI.

REFERENCES

- Atkins, J., S. Mazzi and C. Easter (2000), 'A Commonwealth Vulnerability Index for Developing Countries', London: Commonwealth Secretariat, *Economic Paper* 40.
- Bhinda, N., Griffith-Jones, S., and Martin, M. (1999), 'The scale and monitoring of capital flows', in N. Bhinda, S. Griffith-Jones, J. Leape and M. Martin (eds) *Private Capital Flows to Africa: Perception and Reality*, The Hague: Forum on Debt and Development (FONDAD).
- Bleaney, M., N. Gemmell and D. Greenaway (1995), 'Tax Revenue Instability with particular Reference to sub-Saharan Africa', *Journal of Development Studies*, 31:6, 883-902.
- Cnossen, T. and O. Morrissey (1997), 'Aid and tax Instability in sub-Saharan Africa', paper presented at the *ESRC-DESG Conference on Financing Development*, University of Birmingham, 7-8 September 1997.
- DAC (1999), *Development Co-operation 1998 Report*, Paris: OECD, Development Assistance Committee (DAC).
- Gabriele, A., Baratav, K. and Parikh, A. (2000), 'Instability and volatility of capital flows to developing countries', *The World Economy*, 23:8, 1030-56.
- Lensink, R. and O. Morrissey (2000), 'Aid Instability as a Measure of Uncertainty and the Positive Impact of Aid on Growth' *Journal of Development Studies*, 36:3, 31-49.
- Pallage, S. and M. Robe (2002), 'Foreign Aid and the Business Cycle', *Review of International Economics*, (forthcoming).

The 29 low income countries (LICs, Table A1a) had a 1998 per capita GNP of no more than \$760. The 19 lower-middle income countries (LMICs, Table A1b) had a 1998 per capita GNP in the range \$760 to \$3,030. The 12 upper-middle income countries (UMICs, Table A1c) had a 1998 per capita GNP in the range \$3,031 to \$9,360. All capital inflows are measured in \$US and expressed as a per cent of GDP (the implicit deflator) measured in \$US. Pallage and Robe (2002) show that their measures of aid instability are robust to the use of purchasing power parity measures of GDP. There is no reason to assume that alternative deflators would have significant effects on our measures of instability.

The data on the aid variables are from the *Geographical Distribution of Financial Flows* 1999 (DAC, CD-ROM); FDI, private flows and debt data are from *World Development Indicators 2000* (World Bank, CD-ROM). Variables defined as:

- *Official Development Assistance (ODA)*, net and gross disbursements. ODA disbursements include grants and loans by the official sector at concessional financial terms with the promotion of economic development and welfare as the principal objective. It includes technical co-operation but excludes grants, loans or credits for military purposes. For a definition see DAC (1999: 131).
- *Official Development Finance (ODF)*, net disbursements. ODF disbursements include ODA plus non-concessional resources from multilateral and bilateral sources (including refinancing loans). For a definition see DAC (1999: 131).
- *Foreign Direct Investment (FDI)*, net inflows of investment with the aim of acquiring a lasting management interest in an economy other than that of the investor. Includes equity capital and reinvestment of earnings.
- *Other Private Capital (OPC)* consists of commercial bank lending, bonds, other private credit, non-debt flows, portfolio equity investments and FDI. We subtract FDI from the total private capital flows to get OPC.
- *Total external debt* (stock) is defined as all debt owed to non-residents that is repayable in foreign currency, goods or services. It includes long term debt (public, public guaranteed and private non-guaranteed long term debt and the use of IMF credit) and short term debt (consisting of all debt having an original maturity of one year or less and interest in arrears of long term debt). We obtain the external debt flows as the first difference of the stock of debt.

Most empirical studies have used a measure of the total amount of instability for the variable over a given sample period for each country (e.g. Pallage and Robe, 2002). We note a couple of drawbacks with the 'single number' instability measure. First, this single instability measure means one cannot exploit the annual variability in the series. Second, the averaging of positive (when actual flows exceed the trend) and negative (when actual flows falls short of the trend) instability may mask the trend in instability. Furthermore, there is no reason why negative and positive instability would have the same economic effects (although we do not relate instability of inflows to economic performance in this paper). We therefore estimated 'annual instability' based on an abstraction from the normalised deviations from a linear trend (or forecast value), although we do not report the results.

To do this we first calculate the indices recursively starting with the first three years. The difference between successive recursive indices can be considered as the additional instability that results from increasing the sample period by one observation unit. Therefore the first difference of the recursive averages will be a good approximation of

the index for each year (starting from year 4 in our case). This index will be a measure of the annual total instability as we use fluctuations of the observed values from a trend. Regarding its interpretation, a negative (positive) instability value does not mean that the variable in itself was lower (higher) than some expected value. It merely suggests that instability has reduced (increased) in that year compared to previous years.

A second method of obtaining annual series is to estimate a rolling series. For the start period one estimates a normalised standard deviation for the *n* series starting from say year *t*; for the second period one estimates the standard deviation for the *n* series starting from year t+1 and so on. A potential problem with this approach is that observations for years that have a very high standard deviation could dominate the indices for overlapping years. Nevertheless, this provides an annual series to compare with the first approach. This is useful as we have no other benchmark for comparison.

Given that we estimate instability in the various capital flows for each country it is important that we determine the time series properties of the data. This is done using the augmented Dickey Fuller test (details on request). Our results show that of the 29 LICs in our sample, external debt to GDP is stationary for 9 of them; FDI, OPC, ODA and ODF are stationary for 16, 17, 13, and 9 respectively. For the LMICs we find that 1, 8, 9, 8, and 6 of the total of 19 countries are stationary in external debt, FDI, OPC, ODA and ODF respectively. Finally, of the 12 UMICs, 2, 4, 6, 5, and 3 are stationary in external debt, FDI, OPC, ODA, and ODF respectively. This suggests that less then 40 per cent of all the series are stationary. However we proceed with the estimation of the instability indices on the assumption that the series are integrated of order one. By so doing we sacrifice efficiency for consistency.

	ODA			ODF			
Country	CoV	Index I	Index II	CoV	Index I	Index II	
Benin	41.44	24.93	24.59	43.67	28.08	25.68	
Burkina Faso	33.07	20.66	17.59	32.21	20.20	17.32	
Cameroon	41.09	40.56	36.40	39.25	39.01	36.47	
Central African Rep.	25.74	22.47	26.57	26.17	22.80	26.18	
Chad	40.78	23.80	24.57	40.56	23.73	24.52	
Congo, Dem. Rep.	56.91	53.13	42.94	74.62	72.45	70.73	
Congo, Rep.	56.21	53.93	55.78	52.68	50.54	55.73	
Cote d'Ivoire	95.77	78.27	71.94	69.68	57.18	54.20	
Gambia	60.20	55.47	44.07	61.04	57.05	45.37	
Ghana	61.81	35.17	31.49	62.64	36.01	22.83	
Haiti	78.60	65.30	64.52	78.66	65.22	64.46	
Honduras	51.30	28.37	33.12	44.74	32.35	34.70	
India	36.78	24.07	24.53	29.04	25.71	28.06	
Indonesia	78.23	57.89	19.18	62.58	55.10	33.39	
Kenya	51.06	38.80	38.09	37.41	33.42	33.34	
Madagascar	62.96	38.34	43.05	63.51	41.48	49.85	
Malawi	53.40	34.57	33.18	48.75	32.26	32.22	
Mali	33.64	26.84	24.65	33.50	26.25	23.59	
Mauritania	40.22	36.71	36.56	43.87	41.13	37.32	
Nicaragua	115.33	76.21	75.79	122.62	84.25	80.07	
Niger	39.43	25.92	28.68	37.02	27.37	29.55	
Nigeria	82.44	80.18	57.52	117.89	114.77	70.64	
Pakistan	42.84	30.45	34.27	37.39	32.13	33.91	
Rwanda	96.56	83.95	89.49	96.33	83.84	89.36	
Senegal	34.19	25.54	28.12	34.77	28.75	27.41	
Sierra Leone	89.15	50.82	33.57	84.25	49.41	34.21	
Solomon Islands	39.31	31.13	32.40	38.16	29.98	29.97	
Sudan	51.90	50.69	42.00	54.45	54.29	42.40	
Togo	32.99	27.25	26.74	36.38	34.64	29.20	

Table A1a: Instability in Official Flows – LICs

	ODA				ODF	
Country	CoV	Index I	Index II	CoV	Index I	Index II
Algeria	80.81	70.09	26.89	114.74	109.28	73.14
Bolivia	41.27	15.67	18.83	39.53	21.89	20.29
Colombia	106.70	79.97	33.74	83.26	50.93	29.01
Costa Rica	83.15	84.56	55.80	72.09	73.45	45.17
Dominican Rep.	58.99	58.38	57.64	68.54	69.78	71.93
Ecuador	40.78	40.68	30.56	60.83	56.86	52.35
Egypt, Arab Rep.	68.29	67.36	47.02	63.36	62.71	49.31
El Salvador	67.70	59.56	28.79	62.78	52.96	28.57
Guatemala	49.84	44.00	30.65	35.77	32.18	27.27
Jamaica	63.20	60.80	31.10	75.38	75.05	47.66
Jordan	58.55	51.89	51.07	53.55	49.99	44.93
Morocco	47.71	48.41	54.88	52.35	53.33	42.14
Paraguay	40.91	34.35	27.19	38.20	26.92	26.97
Peru	39.73	38.01	38.89	56.49	45.09	44.80
Philippines	47.15	44.60	39.43	46.34	43.81	40.83
Sri Lanka	47.17	46.39	31.55	46.76	46.12	30.63
Swaziland	40.95	39.66	29.27	53.36	51.23	33.42
Thailand	33.98	32.76	22.89	49.88	49.70	38.93
Tunisia	55.30	26.45	18.67	37.99	29.77	26.48

 Table A1b: Instability in Official Flows – LMICs

		ODA			ODF	
Country	CoV	Index I	Index II	CoV	Index I	Index II
Barbados	82.20	42.33	39.43	68.98	59.26	56.21
Botswana	54.70	14.14	10.00	94.22	70.32	35.46
Brazil	104.94	80.31	47.01	70.58	62.56	56.21
Chile	136.84	136.06	152.99	152.89	152.09	149.32
Gabon	65.23	60.86	45.84	89.88	84.54	85.47
Malaysia	69.20	56.62	51.54	65.32	45.41	50.45
Mauritius	45.30	33.74	29.54	57.32	49.52	34.24
Mexico	54.05	54.28	48.21	93.05	94.78	103.42
Panama	44.46	39.83	36.27	58.72	52.72	51.12
Trinidad&Tobago	83.67	84.94	76.54	89.82	76.79	72.40
Turkey	81.58	76.24	68.60	84.86	76.56	57.87
Venezuela	90.30	90.92	77.84	219.41	211.14	149.63

 Table A1c:
 Instability in Official Flows – UMICs

	FDI			OPC		
Country	CoV	Index I	Index II	CoV	Index I	Index II
Benin	135.99	138.12	110.69	309.93	313.25	341.76
Burkina Faso	158.65	148.96	128.61	389.57	391.50	401.68
Cameroon	152.25	149.41	148.88	150.46	138.32	140.02
Central African Rep.	149.58	140.96	160.22	326.97	304.94	193.35
Chad	115.66	117.82	130.18	228.32	203.36	173.55
Congo, Dem. Rep.	987.04	989.89	992.31	169.39	119.90	66.01
Congo, Rep.	178.25	146.88	45.05	205.97	205.17	236.53
Cote d'Ivoire	104.42	106.34	97.43	150.64	124.22	105.06
Gambia, The	105.80	105.06	121.38	616.49	604.72	634.38
Ghana	136.89	138.09	133.01	265.94	232.80	255.82
Haiti	81.33	47.36	49.90	476.33	482.80	585.76
Honduras	66.10	51.67	50.08	127.11	124.29	134.49
India	163.82	131.26	57.09	97.85	63.93	72.28
Indonesia	82.13	74.98	41.42	83.18	82.32	72.24
Kenya	94.04	89.38	89.28	238.68	212.00	214.40
Madagascar	114.87	116.86	81.69	339.29	330.53	303.81
Malawi	159.82	105.46	93.22	594.05	543.79	362.85
Mali	240.58	226.91	238.08	238.37	224.17	265.55
Mauritania	2529.11	2553.47	3279.19	495.72	503.04	579.52
Nicaragua	172.15	159.34	54.01	270.44	227.46	221.22
Niger	225.20	223.82	267.44	427.29	411.29	423.16
Nigeria	91.36	78.35	80.25	2395.25	2345.29	2283.09
Pakistan	89.97	46.71	35.11	155.10	131.27	119.64
Rwanda	68.30	65.81	42.86	597.13	580.25	578.94
Senegal	148.79	151.60	166.51	227.16	182.38	176.57
Sierra Leone	2215.42	2229.77	2847.11	457.91	436.43	473.76
Solomon Islands	97.70	95.78	108.12	777.89	785.19	728.05
Sudan	504.25	484.21	570.55	188.45	169.32	150.80
Togo	291.36	294.21	339.81	250.75	237.63	142.55

 Table A2a:
 Instability in Private Flows – LICs

		FDI			OPC	
Country	CoV	Index I	Index II	CoV	Index I	Index II
Algeria	411.79	412.33	272.49	151.11	121.23	116.47
Bolivia	217.23	177.36	116.59	184.49	172.45	118.12
Colombia	101.14	75.05	69.58	111.80	104.39	94.30
Costa Rica	41.92	36.19	27.02	155.63	112.27	98.44
Dominican Rep.	61.94	62.66	42.07	198.63	143.23	141.10
Ecuador	114.24	113.69	47.24	121.57	112.47	111.30
Egypt, Arab Rep.	91.71	90.42	90.97	133.90	135.44	97.64
El Salvador	90.40	83.60	100.66	660.62	620.38	637.56
Guatemala	67.26	66.04	80.83	273.85	259.88	252.18
Jamaica	145.21	147.12	64.97	-9834.30	-10010.98	-11643.70
Jordan	153.62	155.31	172.26	190.27	193.44	157.78
Morocco	114.58	90.19	77.17	115.47	109.34	101.32
Paraguay	75.46	69.65	42.60	222.58	197.72	188.95
Peru	188.88	160.16	140.41	120.78	122.78	123.26
Philippines	109.74	76.98	87.94	121.16	122.97	109.32
Sri Lanka	102.72	79.89	80.74	307.65	313.22	290.69
Swaziland	89.82	85.87	85.25	-1088.87	-1101.50	-1343.45
Thailand	69.87	55.21	54.05	88.57	79.50	98.11
Tunisia	59.25	60.10	55.57	226.32	229.76	227.85

 Table A2b:
 Instability in Private Flows – LMICs

		FDI			OPC	
Country	CoV	Index I	Index II	CoV	Index I	Index II
Barbados	137.70	109.32	75.14	261.34	262.60	280.89
Botswana	214.62	218.45	225.67	409.60	405.17	428.84
Brazil	56.02	55.38	43.54	73.16	64.36	54.29
Chile	165.19	100.30	117.34	69.78	70.72	58.47
Gabon	172.57	135.38	148.02	204.52	171.23	150.32
Malaysia	50.00	42.85	35.60	148.46	150.85	130.74
Mauritius	84.47	73.14	61.17	232.49	210.93	212.31
Mexico	66.46	49.26	45.44	82.82	83.60	69.94
Panama	167.32	154.10	139.46	127.33	110.45	113.68
Trinidad&Tobago	84.08	85.61	60.75	1499.48	1388.51	1611.43
Turkey	68.63	55.47	44.53	123.31	100.95	105.58
Venezuela	290.67	228.33	239.76	244.00	240.09	216.92

 Table A2c:
 Instability in Private Flows – UMICs

	External Debt (stock)			External Debt (flows)			
Country	CoV	Index I	Index II	CoV	Index I	Index II	
Benin	54.90	23.86	21.75	115.11	117.22	128.69	
Burkina Faso	59.66	17.09	13.87	109.33	111.61	136.40	
Cameroon	62.87	29.44	22.13	92.65	94.37	107.45	
Central African Rep.	55.62	20.90	21.20	98.33	96.75	102.12	
Chad	57.97	29.09	18.34	126.80	124.82	121.33	
Congo, Dem. Rep.	77.58	25.77	18.28	124.43	126.79	132.75	
Congo, Rep.	58.98	22.84	20.00	138.09	137.98	172.18	
Cote d'Ivoire	53.86	17.00	15.53	187.68	179.96	199.78	
Gambia, The	59.79	31.70	17.13	101.44	101.74	114.36	
Ghana	47.01	14.48	12.21	93.81	92.09	109.14	
Haiti	43.61	26.39	20.46	124.63	124.39	124.41	
Honduras	47.36	12.38	11.62	82.17	80.43	97.81	
India	38.58	22.36	10.30	90.54	92.33	91.65	
Indonesia	29.60	21.66	12.92	58.33	60.92	66.90	
Kenya	41.95	24.02	26.64	130.18	119.94	142.59	
Madagascar	58.61	31.97	24.57	283.02	288.90	335.61	
Malawi	39.48	23.56	26.09	86.77	88.32	111.46	
Mali	38.99	20.18	17.62	105.26	107.52	119.72	
Mauritania	51.13	14.29	11.44	90.34	86.64	93.97	
Nicaragua	102.22	71.55	46.80	584.91	578.30	526.81	
Niger	59.35	21.69	18.90	214.55	213.98	270.50	
Nigeria	82.29	43.70	32.64	212.96	216.96	217.76	
Pakistan	15.14	15.43	8.12	64.12	58.22	55.18	
Rwanda	100.83	59.39	64.86	65.20	64.32	70.62	
Senegal	47.62	28.73	17.32	145.85	140.20	161.20	
Sierra Leone	65.32	29.40	22.20	227.64	229.73	279.74	
Solomon Islands	65.13	33.59	30.66	187.49	191.17	200.49	
Sudan	77.33	37.92	33.88	84.57	80.24	81.04	
Togo	47.68	35.35	24.29	232.14	227.73	246.87	

Table A3a: Instability in External Debt – LICs

	External Debt			Extern		
~	(ste	ock)		 (fl	ow)	
Country	CoV	Index I	Index II	CoV	Index I	Index II
Algeria	34.03	23.39	18.49	156.10	139.52	120.20
Bolivia	32.32	32.81	13.27	151.67	140.65	156.93
Colombia	23.05	20.49	9.18	67.84	69.08	76.56
Costa Rica	51.37	51.76	25.44	161.47	135.00	140.15
Dominican Rep.	44.61	37.07	23.33	100.13	88.73	92.79
Ecuador	49.05	26.65	14.73	105.81	103.29	102.02
Egypt, Arab Rep.	43.58	14.17	19.33	191.46	172.67	195.90
El Salvador	34.59	93.54	11.09	107.30	104.65	110.12
Guatemala	48.16	31.60	19.01	100.30	100.96	120.93
Jamaica	39.44	33.03	19.32	199.35	192.30	188.64
Jordan	68.08	39.27	26.87	164.67	167.14	179.12
Morocco	45.00	35.81	15.32	123.74	114.17	123.89
Paraguay	52.64	49.94	20.81	179.80	172.35	177.58
Peru	29.60	28.61	27.85	97.81	98.63	113.50
Philippines	34.30	28.42	10.17	86.64	82.59	83.93
Sri Lanka	36.87	20.68	12.91	79.37	80.35	77.51
Swaziland	45.13	45.83	29.50	242.77	245.21	291.57
Thailand	43.32	17.06	12.42	59.12	54.62	61.30
Tunisia	26.52	15.22	9.87	94.21	94.17	99.90

 Table A3b:
 Instability in External Debt – LMICs

	Exter	nal debt (stock)	External debt (flows)			
Country	CoV	Index I	Index II	CoV	Index I	Index II	
Barbados	51.49	35.98	26.97	271.36	270.64	303.33	
Botswana	46.16	31.59	20.92	193.66	168.07	138.89	
Brazil	31.04	30.31	15.61	72.29	53.17	53.89	
Chile	43.76	44.22	21.24	119.98	115.95	120.96	
Gabon	38.23	27.75	21.64	144.17	137.87	119.02	
Malaysia	46.21	39.21	16.92	119.80	121.72	111.65	
Mauritius	47.31	35.38	17.07	107.44	108.03	121.14	
Mexico	42.29	38.16	22.91	112.25	101.97	115.95	
Panama	32.93	25.56	10.74	103.25	76.88	93.56	
Trinidad&Tobago	63.33	29.45	21.31	231.84	231.38	263.99	
Turkey	40.16	18.98	16.69	92.95	93.70	118.63	
Venezuela	47.86	30.02	19.93	208.88	193.23	190.74	

 Table A3c:
 Instability in External Debt – UMICs

	LICs	LMICs	UMICs
1970 – 1975	6.83	3.56	2.63
1976 – 1980	8.73	4.71	1.86
1981 – 1985	9.42	4.33	1.63
1986 – 1990	12.97	4.73	1.42
1991 – 1995	16.08	4.61	1.08
1996 – 1997	13.45	2.74	0.89
Entire period	10.85	4.24	1.70
St. Deviation	3.64	0.97	0.63
Minimum	5.37	2.53	0.87
Maximum	20.84	7.27	3.21

Table A4a: Trends in Gross ODA to GDP Ratio

Table A4b: Instability in Gross ODA across countries

OVERALL	LICs	LMICs	UMICs
CoV	33.51	22.97	36.88
Index 1	15.28	23.27	15.24
Index 2	15.72	20.54	13.87
TRENDS (annual avera	ige)		
1972 – 1980	-1.02	-5.05	-1.05
1981 – 1989	0.08	-0.68	-0.29
1990 – 1997	0.75	1.07	-0.05
Period	-0.06	-1.52	-0.45

Notes: Estimates based on average flows to countries in each group. Trends based on annual instability measures obtained as the first difference of the recursive estimates for Index I. Period is annual average over 1970-97.

	Debt (stock)	Debt (flows)	FDI	OPC	ODA	ODF	
			LIC	Ĉs -			
1972 – 1980	-5.14	-15.75	-2.26	-38.09	0.09	-0.87	
1981 – 1989	-1.04	1.19	1.99	1.02	0.004	-0.20	
1990 - 1997	0.25	3.21	-0.49	1.90	0.76	0.76	
Period	-1.81	-2.53	-0.16	-11.21	0.27	-0.11	
	LMICs						
1972 - 1980	-3.41	-24.13	-4.52	-29.25	-7.43	-8.07	
1981 – 1989	-0.05	-0.18	1.05	-0.57	-0.81	-0.79	
1990 - 1997	-0.10	1.18	-0.80	1.65	0.57	0.27	
Period	-1.04	-5.95	-1.33	-9.04	-2.49	-2.78	
			UMI	Cs			
1972 - 1980	-3.43	-1.06	-4.43	-5.69	-0.81	-3.89	
1981 – 1989	0.43	1.98	0.54	1.39	-0.26	-0.95	
1990 - 1997	-0.26	0.53	0.49	1.02	0.05	0.04	
Period	-0.92	0.68	-1.07	-1.00	-0.34	-1.57	

Table A5: Instability of Capital Flows using index I (annual average trend)

Notes: Estimates based on average flows to countries in each group. Annual instability measures are obtained as the first difference of the recursive estimates for Index I. Period is annual average over 1970-97.

Table A6: Instability of Capital Flows using CoV (entire period, all countries)

	Debt	Debt	FDI	OPC	ODA	ODF
	(stock)	(flows)				
			LI	Cs		
1970 - 1979	26.84	44.87	26.84	52.25	19.45	20.85
1980 - 1989	28.36	63.84	61.79	121.71	15.50	13.00
1990 - 1997	10.33	272.00	39.23	629.59	17.18	16.42
			LM	ICs		
1970 - 1979	25.32	55.33	22.93	55.56	24.89	26.69
1980 - 1989	17.81	61.02	35.92	49.12	6.01	6.76
1990 - 1997	12.64	140.94	27.24	164.44	37.62	27.87
	UMICs					
1970 - 1979	22.24	20.66	30.69	32.66	23.88	46.00
1980 - 1989	20.06	72.31	25.35	77.66	14.39	14.45
1990 – 1997	5.61	72.56	48.37	82.54	32.24	36.37

Notes: Estimates based on average flows to the countries in each group. Instability measures are defined in Section 2.

CREDIT PAPERS

- 00/1 **Robert Lensink,** "Does Financial Development Mitigate Negative Effects of Policy Uncertainty on Economic Growth?"
- 00/2 **Oliver Morrissey,** "Investment and Competition Policy in Developing Countries: Implications of and for the WTO"
- 00/3 Jo-Ann Crawford and Sam Laird, "Regional Trade Agreements and the WTO"
- 00/4 **Sam Laird,** "Multilateral Market Access Negotiations in Goods and Services"
- 00/5 Sam Laird, "The WTO Agenda and the Developing Countries"
- 00/6 Josaphat P. Kweka and Oliver Morrissey, "Government Spending and Economic Growth in Tanzania, 1965-1996"
- 00/7 Henrik Hansen and Fin Tarp, "Aid and Growth Regressions"
- 00/8 Andrew McKay, Chris Milner and Oliver Morrissey, "The Trade and Welfare Effects of a Regional Economic Partnership Agreement"
- 00/9 Mark McGillivray and Oliver Morrissey, "Aid Illusion and Public Sector Fiscal Behaviour"
- 00/10 **C.W. Morgan,** "Commodity Futures Markets in LDCs: A Review and Prospects"
- 00/11 Michael Bleaney and Akira Nishiyama, "Explaining Growth: A Contest between Models"
- 00/12 **Christophe Muller,** "Do Agricultural Outputs of Autarkic Peasants Affect Their Health and Nutrition? Evidence from Rwanda"
- 00/13 **Paula K. Lorgelly,** "Are There Gender-Separate Human Capital Effects on Growth? A Review of the Recent Empirical Literature"
- 00/14 **Stephen Knowles and Arlene Garces,** "Measuring Government Intervention and Estimating its Effect on Output: With Reference to the High Performing Asian Economies"
- 00/15 I. Dasgupta, R. Palmer-Jones and A. Parikh, "Between Cultures and Markets: An Eclectic Analysis of Juvenile Gender Ratios in India"
- 00/16 **Sam Laird,** "Dolphins, Turtles, Mad Cows and Butterflies A Look at the Multilateral Trading System in the 21st Century"
- 00/17 Carl-Johan Dalgaard and Henrik Hansen, "On Aid, Growth, and Good Policies"
- 01/01 **Tim Lloyd, Oliver Morrissey and Robert Osei,** "Aid, Exports and Growth in Ghana"
- 01/02 **Christophe Muller,** "Relative Poverty from the Perspective of Social Class: Evidence from The Netherlands"
- 01/03 **Stephen Knowles,** "Inequality and Economic Growth: The Empirical Relationship Reconsidered in the Light of Comparable Data"
- 01/04 A. Cuadros, V. Orts and M.T. Alguacil, "Openness and Growth: Re-Examining Foreign Direct Investment and Output Linkages in Latin America"
- 01/05 Harold Alderman, Simon Appleton, Lawrence Haddad, Lina Song and Yisehac Yohannes, "Reducing Child Malnutrition: How Far Does Income Growth Take Us?"

- 01/06 **Robert Lensink and Oliver Morrissey,** "Foreign Direct Investment: Flows, Volatility and Growth"
- 01/07 Adam Blake, Andrew McKay and Oliver Morrissey, "The Impact on Uganda of Agricultural Trade Liberalisation"
- 01/08 **R. Quentin Grafton, Stephen Knowles and P. Dorian Owen,** "Social Divergence and Economic Performance"
- 01/09 **David Byrne and Eric Strobl,** "Defining Unemployment in Developing Countries: The Case of Trinidad and Tobago"
- 01/10 **Holger Görg and Eric Strobl,** "The Incidence of Visible Underemployment: Evidence for Trinidad and Tobago"
- 01/11 **Abbi Mamo Kedir,** "Some Issues in Using Unit Values as Prices in the Estimation of Own-Price Elasticities: Evidence from Urban Ethiopia"
- 01/12 Eric Strobl and Frank Walsh, "Minimum Wages and Compliance: The Case of Trinidad and Tobago"
- 01/13 Mark McGillivray and Oliver Morrissey, "A Review of Evidence on the Fiscal Effects of Aid"
- 01/14 **Tim Lloyd, Oliver Morrissey and Robert Osei,** "Problems with Pooling in Panel Data Analysis for Developing Countries: The Case of Aid and Trade Relationships"
- 01/15 **Oliver Morrissey,** "Pro-Poor Conditionality for Aid and Debt Relief in East Africa"
- 01/16 Zdenek Drabek and Sam Laird, "Can Trade Policy help Mobilize Financial Resources for Economic Development?"
- 01/17 **Michael Bleaney and Lisenda Lisenda**, "Monetary Policy After Financial Liberalisation: A Central Bank Reaction Function for Botswana"
- 01/18 **Holger Görg and Eric Strobl,** "Relative Wages, Openness and Skill-Biased Technological Change in Ghana"
- 01/19 **Dirk Willem te Velde and Oliver Morrissey,** "Foreign Ownership and Wages: Evidence from Five African Countries"
- 01/20 **Suleiman Abrar,** "Duality, Choice of Functional Form and Peasant Supply Response in Ethiopia"
- 01/21 **John Rand and Finn Tarp,** "Business Cycles in Developing Countries: Are They Different?"
- 01/22 Simon Appleton, "Education, Incomes and Poverty in Uganda in the 1990s"
- 02/01 Eric Strobl and Robert Thornton, "Do Large Employers Pay More in Developing Countries? The Case of Five African Countries"
- 02/02 Mark McGillivray and J. Ram Pillarisetti, "International Inequality in Human Development, Real Income and Gender-related Development"
- 02/03 **Sourafel Girma and Abbi M. Kedir,** "When Does Food Stop Being a Luxury? Evidence from Quadratic Engel Curves with Measurement Error"
- 02/04 Indraneel Dasgupta and Ravi Kanbur, "Class, Community, Inequality"
- 02/05 Karuna Gomanee, Sourafel Girma and Oliver Morrissey, "Aid and Growth in Sub-Saharan Africa: Accounting for Transmission Mechanisms"
- 02/06 **Michael Bleaney and Marco Gunderman,** "Stabilisations, Crises and the "Exit" Problem A Theoretical Model"

- 02/07 Eric Strobl and Frank Walsh, "Getting It Right: Employment Subsidy or Minimum Wage? Evidence from Trinidad and Tobago"
- 02/08 Carl-Johan Dalgaard, Henrik Hansen and Finn Tarp, "On the Empirics of Foreign Aid and Growth"
- 02/09 **Teresa Alguacil, Ana Cuadros and Vincente Orts,** "Does Saving Really Matter for Growth? Mexico (1970-2000)"
- 02/10 Simon Feeny and Mark McGillivray, "Modelling Inter-temporal Aid Allocation"
- 02/11 Mark McGillivray, "Aid, Economic Reform and Public Sector Fiscal Behaviour in Developing Countries"
- 02/12 **Indraneel Dasgupta and Ravi Kanbur,** "How Workers Get Poor *Because* Capitalists Get Rich: A General Equilibrium Model of Labor Supply, Community, and the Class Distribution of Income"
- 02/13 Lucian Cernat, Sam Laird and Alessandro Turrini, "How Important are Market Access Issues for Developing Countries in the Doha Agenda?"
- 02/14 Ravi Kanbur, "Education, Empowerment and Gender Inequalities"
- 02/15 **Eric Strobl,** "Is Education Used as a Signaling Device for Productivity in Developing Countries?"
- 02/16 Suleiman Abrar, Oliver Morrissey and Tony Rayner, "Supply Response of Peasant Farmers in Ethiopia"
- 02/17 Stephen Knowles, "Does Social Capital Affect Foreign Aid Allocations?"
- 02/18 **Dirk Willem te Velde and Oliver Morrissey,** "Spatial Inequality for Manufacturing Wages in Five African Countries"
- 02/19 Jennifer Mbabazi, Oliver Morrissey and Chris Milner, "The Fragility of the Evidence on Inequality, Trade Liberalisation, Growth and Poverty"
- 02/20 **Robert Osei, Oliver Morrissey and Robert Lensink,** "The Volatility of Capital Inflows: Measures and Trends for Developing Countries"

SCHOOL OF ECONOMICS DISCUSSION PAPERS

In addition to the CREDIT series of research papers the School of Economics produces a discussion paper series dealing with more general aspects of economics. Below is a list of recent titles published in this series.

- 00/1 Tae-Hwan Kim and Christophe Muller, "Two-Stage Quantile Regression"
- 00/2 Spiros Bougheas, Panicos O. Demetrides and Edgar L.W. Morgenroth, "International Aspects of Public Infrastructure Investment"
- 00/3 Michael Bleaney, "Inflation as Taxation: Theory and Evidence"
- 00/4 Michael Bleaney, "Financial Fragility and Currency Crises"
- 00/5 **Sourafel Girma,** "A Quasi-Differencing Approach to Dynamic Modelling from a Time Series of Independent Cross Sections"
- 00/6 **Spiros Bougheas and Paul Downward,** "The Economics of Professional Sports Leagues: A Bargaining Approach"
- 00/7 Marta Aloi, Hans Jørgen Jacobsen and Teresa Lloyd-Braga, "Endogenous Business Cycles and Stabilization Policies"
- 00/8 A. Ghoshray, T.A. Lloyd and A.J. Rayner, "EU Wheat Prices and its Relation with Other Major Wheat Export Prices"
- 00/9 **Christophe Muller**, "Transient-Seasonal and Chronic Poverty of Peasants: Evidence from Rwanda"
- 00/10 **Gwendolyn C. Morrison**, "Embedding and Substitution in Willingness to Pay"
- 00/11 **Claudio Zoli,** "Inverse Sequential Stochastic Dominance: Rank-Dependent Welfare, Deprivation and Poverty Measurement"
- 00/12 **Tae-Hwan Kim, Stephen Leybourne and Paul Newbold,** "Unit Root Tests With a Break in Variance"
- 00/13 **Tae-Hwan Kim, Stephen Leybourne and Paul Newbold**, "Asymptotic Mean Squared Forecast Error When an Autoregression With Linear Trend is Fitted to Data Generated by an I(0) or I(1) Process"
- 00/14 **Michelle Haynes and Steve Thompson,** "The Productivity Impact of IT Deployment: An Empirical Evaluation of ATM Introduction"
- 00/15 Michelle Haynes, Steve Thompson and Mike Wright, "The Determinants of Corporate Divestment in the UK"
- 00/16 John Beath, Robert Owen, Joanna Poyago-Theotoky and David Ulph, "Optimal Incentives for Incoming Generations within Universities"
- 00/17 S. McCorriston, C. W. Morgan and A. J. Rayner, "Price Transmission: The Interaction Between Firm Behaviour and Returns to Scale"
- 00/18 **Tae-Hwan Kim, Douglas Stone and Halbert White,** "Asymptotic and Bayesian Confidence Intervals for Sharpe Style Weights"
- 00/19 **Tae-Hwan Kim and Halbert White,** "James-Stein Type Estimators in Large Samples with Application to the Least Absolute Deviation Estimator"
- 00/20 **Gwendolyn C. Morrison,** "Expected Utility and the Endowment Effect: Some Experimental Results"
- 00/21 **Christophe Muller**, "Price Index Distribution and Utilitarian Social Evaluation Functions"

- 00/22 **Michael Bleaney,** "Investor Sentiment, Discounts and Returns on Closed-End Funds"
- 00/23 **Richard Cornes and Roger Hartley,** "Joint Production Games and Share Functions"
- 00/24 **Joanna Poyago-Theotoky,** "Voluntary Approaches, Emission Taxation and the Organization of Environmental R&D"
- 00/25 Michael Bleaney, Norman Gemmell and Richard Kneller, "Testing the Endogenous Growth Model: Public Expenditure, Taxation and Growth Over the Long-Run"
- 00/26 **Michael Bleaney and Marco Gundermann,** "Credibility Gains and Output Losses: A Model of Exchange Rate Anchors"
- 00/27 **Indraneel Dasgupta,** "Gender Biased Redistribution and Intra-Household Distribution"
- 00/28 **Richard Cornes and Roger Hartley,** "Rentseeking by Players with Constant Absolute Risk Aversion"
- 00/29 S.J. Leybourne, P. Newbold, D. Vougas and T. Kim, "A Direct Test for Cointegration Between a Pair of Time Series"
- 00/30 **Claudio Zoli,** "Inverse Stochastic Dominance, Inequality Measurement and Gini Indices"
- 01/01 Spiros Bougheas, "Optimism, Education, and Industrial Development"
- 01/02 **Tae-Hwan Kim and Paul Newbold,** "Unit Root Tests Based on Inequality-Restricted Estimators"
- 01/03 Christophe Muller, "Defining Poverty Lines as a Fraction of Central Tendency"
- 01/04 **Claudio Piga and Joanna Poyago-Theotoky**, "Shall We Meet Halfway? Endogenous Spillovers and Locational Choice"
- 01/05 **Ilias Skamnelos,** "Sunspot Panics, Information-Based Bank Runs and Suspension of Deposit Convertibility"
- 01/06 **Spiros Bougheas and Yannis Georgellis,** "Apprenticeship Training, Earnings Profiles and Labour Turnover: Theory and German Evidence"
- 01/07 **M.J. Andrews, S. Bradley and R. Upward,** "Employer Search, Vacancy Duration and Skill Shortages"
- 01/08 Marta Aloi and Laurence Lasselle, "Growing Through Subsidies"
- 01/09 **Marta Aloi and Huw D. Dixon,** "Entry Dynamics, Capacity Utilisation, and Productivity in a Dynamic Open Economy"
- 01/10 **Richard Cornes and Roger Hartley,** "Asymmetric Contests with General Technologies"
- 01/11 Richard Cornes and Roger Hartley, "Disguised Aggregative Games"
- 01/12 Spiros Bougheas and Tim Worrall, "Cost Padding in Regulated Monopolies"
- 10/13 Alan Duncan, Gillian Paull and Jayne Taylor, "Price and Quality in the UK Childcare Market"
- 01/14 **John Creedy and Alan Duncan**, "Aggregating Labour Supply and Feedback Effects in Microsimulation"

Members of the Centre

Director

Oliver Morrissey - aid policy, trade and agriculture

Research Fellows (Internal)

Simon Appleton – poverty, education, household economics Adam Blake – CGE models of low-income countries Mike Bleaney - growth, international macroeconomics Indraneel Dasgupta – development theory, household bargaining Norman Gemmell – growth and public sector issues Ken Ingersent - agricultural trade Tim Lloyd – agricultural commodity markets Andrew McKay - poverty, peasant households, agriculture Chris Milner - trade and development Wyn Morgan - futures markets, commodity markets Christophe Muller – poverty, household panel econometrics Tony Rayner - agricultural policy and trade

Research Fellows (External)

David Fielding (University of Leicester) – investment, monetary and fiscal policy Ravi Kanbur (Cornell) – inequality, public goods – <u>Visiting Research Fellow</u> HenrikHansen (University of Copenhagen) – aid and growth Stephen Knowles (University of Otago) – inequality and growth Sam Laird (UNCTAD) – trade policy, WTO Robert Lensink (University of Groningen) – aid, investment, macroeconomics Scott McDonald (University of Sheffield) – CGE modelling, agriculture Mark McGillivray (WIDER, Helsinki) – aid allocation, aid policy Doug Nelson (Tulane University) - political economy of trade Shelton Nicholls (University of West Indies) – trade, integration Eric Strobl (University College Dublin) – labour markets Finn Tarp (University of Copenhagen) – aid, CGE modelling